Author's response to reviews

Title: Assessing the Validity of Tuberculosis Surveillance Data in California

Authors:

Joan E Sprinson (jsprinso@dhs.ca.gov)
Elizabeth S Lawton (elawton@dhs.ca.gov)
Travis C Porco (tporco@dhs.ca.gov)
Jennifer M Flood (jflood@dhs.ca.gov)
Janice L Westenhouse (jwestenh@dhs.ca.gov)

Version: 3 Date: 10 August 2006

Author's response to reviews: see over
To the Editors:

Thank you for your acceptance of our manuscript, titled Assessing the Validity of Tuberculosis Surveillance Data in California for publication in BMC Public Health. As requested, we have incorporated a statement regarding ethical approval for data collection in the study; please see the second sentence in paragraph two of the Methods section. We have gone through the manuscript formatting checklist, and the manuscript conforms to the journal's formatting requirements. Again, we thank the reviewers for their helpful comments.

Sincerely,

Jennifer Flood, MD, MPH
Tuberculosis Control Branch
Division of Communicable Disease Control
California Department of Health Services
850 Marina Bay Parkway
Richmond, CA 94804